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Linac4 Commissioning Status

IST and Dry-run successful:
v Two special BeamIn/BeamOut timings signals validated

Modulation test on all the 24 channels done

The 2 VME crates now fully operational on LUMENS: reboot sequence tested
Connections to BIC checked (3 CIBU)

Laser Alarms from BLM to CCC deployed

SW Thresholds properly interlocking according on the configuration (CIBU1/CIBU2)
HW Thresholds tested randomly on several channels

NXCALS logging tested

LN X X X X X

The Linac4 BLM system is now fully operated by OP.
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LINAC 4 Vistar and BLM Masking

LINAC 4 Fixed Display
status OK
L4L

113783113
-34.58-26.8

4013
-26.1

96% 100% 99%

AUTOP.
I[set]
I[avg]

e/H-[avg]

Comments: 19-Oct-2020
09:57:13
Coordinator: B. Mikulec
162393

-35.00
-34.84

100% 99% 100%

RF Info
-24.50

0.00
1.23

04/12/2020

99%

26-10-2020 16:08:32
MD5 Dest: LBE
LTB

101% 99% 100% 98%

BIS

Mask SwinterlockBeam

SPARE.1.1
L4D.ELM.0214
L4D.BLM.0308

SPARE. 1.4
L4C.ELM.O114
L4C.ELM.0214
L4C.ELM.0414
L4T.BLM.0134
L4C.ELM.OB14
L4P.ELM.O114
L4P.ELM.04 14
L4P.BLM.1014
L47.BLM.0272
L4T.BLM.0414
L4T.BLM.07 14
L4T.BLM.0818
L4T.BLM. 1084
L4T.BLM.1272
L4T.BLM.1318
L4T.BLM. 1634

LT.BELM20

LT.BELM50

LT.BLM&5

LTB.BLM30

DT

[ L4.BLMA Setting#swinterlockBeamMask(1]

[l Lv4.BLM.A Setting#swinterlockBeamMask(2]

N4.BLM.A/Setting#swinterlockBeamMask[4]
N4.BLM.A/Setting#swinterlockBeamMask[5]
N4.BLM.A/Setting#swinterlockBeamMask[6]
N4.BLM.A/Setting#swinterlockBeamMask[7]
N4.BLM.A/Setting#swinterlockBeamMask[8]
N4.BLM.A/Setting#swinterlockBeamMask[9]
N4.BLM.A/Setting#swinterlockBeamMask[10]
N4.BLM.A/Setting#swinterlockBeamMask[11]

N4.BLM.A/Setting#swinterlockBeamMask([12]

. N4.BLM.A/Setting#swinterlockBeamMask[13]
. LN4.BLM.A/Setting#swinterlockBeamMask([14]
. LN4.BLM.A/Setting#swinterlockBeamMask[15]
. LN4.BLM.A/Setting#swinterlockBeamMask[16]
. LN4.BLM.A/Setting#swinterlockBeamMask([17]
. LN4.BLM.A/Setting#swinterlockBeamMask([18]
. LN4.BLM.A/Setting#swinterlockBeamMask([19]
. LN4.BLM.A/Setting#swinterlockBeamMask[20]
. LN4.BLM.A/Setting#swinterlockBeamMask[21]
. LN4.BLM.A/Setting#swinterlockBeamMask[22]
. LN4.BLM.A/Setting#swinterlockBeamMask([23]

. LBLM.B/Setting#swinterlockBeamMask[0]
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Write LN4.BLM.A/MaskAlISWiInterlocks A

LN4.BLM.A/MaskAlls terlo v

Write LN4.BLM.A/UndoMaskAlISWiInterlocks &

LN4 BLM A/Undo

Write LN4.BLM.B/MaskAlISWInterlocks

nterlocks | true

Courtesy: Yu Wu, BE-OP




BLM Setup with beam in Linac4

Edit View References Archives Commands Control Programs Help
3257 PSE - 25 MD3 | BC 142 2020 — o
BLMIN] Sw Interlock
LINAC4 B X LN4.BLM.A Talse
<) QF my7 _ ¥ BLMIN]_Virtual Loss Beam Presence BLM Wasked BLM Thresholds Sw Dunp Status Bean Presence
LINAC4 BLM Application PSB.USER.MD? 10-19 12 e g o= — — percd
1000 Take screer L4D.BLM.0308 1.3090E-01 false 3.00 false
L4C.BLM.0114 2.4276E01 false 64.52 false
. L4C.BLM.0214 1.2257E00 false 4.00 false
U .BLM.0414 §.8060E-01 Talse 3.00 Talse
1.6333E-01 Talse 3.00 Talse
Exit Inspe 2.1000E-01 Talse z.00 Talse
3.8500E-01 trus 2.00 Talse
8.1667E-02 false 2.00 false
LAD LAC L4P LAT/LAZ LTILTB 0.0000E00 false 50.00 false
0.0000E0D, false 120.00 false
0214 | 0308 0114 0214 0414 0614 0114 0414 1014 0714 0818 1084 1272 1318 1634 2 50 65 0.0000E00 Talsel 100.00 Talsel
0.0000E0D, false 20.30 false
0.0000E0D, false 3.30 false
0.0000E0D, false 1.30 false
0.0000E00D, false 50.00 false
BeamPres 0.0000E00D, false 40.00 false
. 0.0000E0D, false 2.00 false
BLMIN]_Virtual Loss Beam Presence BLM Wasked BLM Thresholds Sw Dunp Status Beam Presence
LT.BLH2O 0.0000EQ0, false 25.00 false
LT.BLHSO 0.0000EQ0, false 25.00 false
LT.BLHES 0.0000EQ0, false 15.00 false
LTB. BLM30 0.0000EQ0 false 30.00 false
LTIM Event Start Delay Clock Str. AgnuTC Agnc Agnchano
BX. SBP-BLMLN4A trus BX.SBP-CT. 0 10MHz 1602613975 0 0
BX. EBP-BLMLN4A false BX.SBP-CT. 0 1kHz - - -
BX. SBIN-BLHLNAA trus EXT1 0 10MHz 1602613975 274 274998300
BX. EBOUT-BLHLN4A trus EXTZ 5 10MHz 1602613975 275 275099400
BLMIN] Sw Interlock
LN4.BLM.B false
BLMIN]_Virtual Loss Beam Presence BLM Wasked BLM Thresholds Sw Dunp Status Beam Presence
LBE. BLM10 0.0000EQ0. false 25.00 false
LTIM Event Start Clock Str. AgnuTC Agnc Agnchano
LN4 BLMs BX. SBP-BLMLN4B trus BX.SBP-CT. 10MHz 1602613975 0 0
BX. EBP-BLMLN4B trus BX.SBP-CT. 1KHz 1602613975 0 0
BX. SBIN-BLHLN4G trus EXT1 10MHz 1602613975 274 274998050
BX. EBOUT-BLMLNAE trus EXT2Z 10MHz 1602613975 275 275099100
Number of Bad Shs 0 Reset Swinterlock / B
4 [*]
InterlockClounter 5 Reset Swinterloc EEEM@EH@@D 15 o o G |
LNABLMA Status LNBLMB_Status | Mo Exception to display...
-

Tuning of SW Interlock Thresholds in LN4 Loss measurement, masks and thresholds in LN4

Courtesy: Yu Wu, BE-OP
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BLM Test Sequence in Linac4

Tog Viewer - PSB.USER.MD7 - (INCAI
M0 4 MOLOL00 VL R SGMOLI BI04 M7 | B e 2020 @] ¢ [wson + = -
: # 4 B 26002020 1953730 FSE_ 16 MO | BC_IRE 2030 = 17MD3 | BC.L4Z 2020 @] [wson =
= , 7
Data aisplay Eait paramete ) Savets csv | ') Load fiom Csv e = :
[~ Duamspray | [T | Ean parameter st Save te CSV Load frem CSV
E-BIC-ROE -k [ ] g Lle
e e | AL : = =
- AERr-RaEc-=2
-
- LADBUMA T4 R cuesinben #los s Beam Fresere
© LALILMA 1 Ao ot 15
- LABBLMA L Acuasiionioa s S 2
o {l canmnniaa )
»
™
»
4
= .
2 PN AN ‘ . / J < 2
15
10
s
%
. ! ' : v . .
1940000 1940200 1940400 1941000 1941200 1942000 1942200 1942400 - B T v v T T
T R — + Mo 1952000 1953000 1954000 1955000 1956000 1557.000 a
- Accumulating values: out o max =
% || @ 0 [idetabie] snow cyctes | show neaser viewes Curser digits 3 400 . .
2 Y @ |« | @ | 5% [Wide table | show cyates | Show Bedset Viewer c it Accumulating values: 3336 out of 100000 max
pector 3.5.28 - LINACA BLM - PSB.USER.MD?
Wi = UNACA BLI x
2410 105 : B — N o
T =T LINAC4 BLM Application PSB.USER.MD7 2020-10-06 19:45:03 iites | 1ray vengs . . _
T A5081 1000000 CIETrr T eea—— e | LINAC4A BLM Application PSB.USER.MD7? 2020-10-06 19:58:41
3.0643E-01 . B L6 BLM 0114 fAequisition#iassAS1 b47SE0L 25E0 oo
1.68986-01 1.4875E-01 4C BLM 0114 AcquishiondiossAs2 5234E01] 3340060
B L4 LM 0314 fAcquisition#iassRsa 5389E01| 2.6573E0
100000
.
I\ LATILAZ LTILTB A LaTiLaz
o Exception to displa .. o8 o 14 Bie 10 ‘ \ o4 osis  lomé 0 s 6 \ LTLTB
i r; No Exception to display. o L oz14 ‘4 614 oM \ oas 0818 084 7 0 65
19752,05 - Spping FIRST updare vatoes and wa
10000

Reset Swinterlack B InterlockCounter

Test of channel L4D.BLM.0214 in LN4 2020-10-06 Test of channel L4D.BLM.0114 in LN4 2020-10-06

Courtesy: Yu Wu, BE-OP
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PSB Commissioning Status

IST and Dry-run successful:

v Modulation test on all the 104 channels done
The 4 VME crates now fully operational on LUMENS: reboot sequence tested

PSB extraction split into 4 virtual instances: BT, BTM, BTY, BTP

Connections to BIC checked (6 CIBU)

Laser Alarms from BLM to CCC deployed

SW Thresholds properly interlocking the Ring1/2 or Ring3/4 according to the configuration

HW Thresholds tested randomly on several channels
NXCALS logging tested

LN X X X X X

The PSB BLM system is commissioned and ready to be used by OP.
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Commissioning Status of the Other Accelerators

PS Ring IST done:

v" Modulation test on all the 100 PS channels done

v" The 2 VME crates now fully operational on LUMENS:
reboot sequence tested

v BLM SYNC concentrator operational
v" NXCALS logging tested

East area IST not started:

» On-going installation of 15 channels (4 already done)
» Modulation test to be done
v" The VME crate now fully operational on LUMENS: reboot sequence tested

04/12/2020 MPP - BLMINJ Commissioning Status 7




Commissioning Status of the Other Accelerators

TT2 IST done:

v" Modulation test on all the 49 channels done
v" The VME crate now fully operational on LUMENS: reboot sequence tested
» Noise peaks in the cables with HV distribution in SWY area (Jira_issue_BIBML-1707 )

TT10 IST done:
v Modulation test on all the 30 channels done
v" The 2 VME crates now fully operational on LUMENS: reboot sequence tested
v' TT10 Timing test mode implemented to see data at BP rate instead of injection rate

BEP Acq. Mode
Status: OFF

""WARNING**
Timing of the crate will be changed!

ENAEBLE
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https://issues.cern.ch/browse/BIBML-1707




